Author’s response to reviews

Title: Mammary phyllodes tumor with six episodes of a relapse: a case report

Authors:
Nozomi Iimori (nozomi_0422_fujita@yahoo.co.jp)
Shinichiro Kashiwagi (spqv9ke9@view.ocn.ne.jp)
Tetsuro Ishikawa (t.ishikawa@city.kashiwara.osaka.jp)
Hidemi Kawajiri (hkawajiri75@yahoo.co.jp)
Tsutomu Takashima (tsutomo-@rd5.so-net.ne.jp)
Masahiko Ohsawa (m-ohsawa@med.osaka-cu.ac.jp)
Kosei Hirakawa (hirakawa@med.osaka-cu.ac.jp)
Masaichi Ohira (masaichi@med.osaka-cu.ac.jp)

Version: 1 Date: 18 Jun 2017

Author’s response to reviews:

Dear Editor-in-Chief

Journal of Medical Case Reports

Michael Kidd, M.D.

Please find our enclosed manuscript, entitled “Mammary phyllodes tumor with six episodes of relapse: a case report” that we are submitting for consideration for publication in Journal of Medical Case Reports. This manuscript is not being considered in whole or in part by other journal at this time. All authors are aware of the consent of this manuscript. We are submitting this paper as an Original Report for Journal of Medical Case Reports.

We confirmed the correction of grammar. We asked native check over the whole paper and proofread an English sentence. (Editage, No. GIZXC_5)
This study examines the case of a phyllodes tumor with 6 episodes of relapse. Pathological analysis confirmed this was a malignant form of phyllodes tumor. We believe that our study makes a significant contribution to the literature because we describe the pathology and clinical course of a rare case of malignant phyllodes tumor with multiple relapses.

Further, we believe that this paper will be of interest to the readership of your journal because this is a rare malignant case of a breast tumor, with an atypical number of relapses. We believe our case report is conforms with the scope and aims of your journal.

We hope you will seriously consider this report for publication in Journal of Medical Case Reports.

Sincerely,

Shinichiro Kashiwagi, M.D., Ph.D.
Osaka City University Graduate School of Medicine,
1-4-3 Asahi-machi, Abeno-ku, Osaka 545-8585, Japan
TEL; (+81) 6-6645-3838
FAX; (+81) 6-6646-6450
e-mail address; spqv9ke9@view.ocn.ne.jp